Author’s response to reviews

Title: Trends in influenza vaccination uptake among people aged over 74 years, 1997-2000: Survey of 73 general practices in Britain

Authors:

Ms Elizabeth Breeze (elizabeth.breeze@lshtm.ac.uk)
Dr Punam Mangtani (punam.mangtani@lshtm.ac.uk)
Gill M Price (gill.price@lshtm.ac.uk)
Sari Kovats (sari.kovats@lshtm.ac.uk)
Jenny Roberts (jenny.roberts@lshtm.ac.uk)
Prof Astrid E Fletcher (astrid.fletcher@lshtm.ac.uk)

Version: 6 Date: 20 Apr 2004

Formatting changed as requested